\documentclass{article}

\usepackage{amsmath,amsfonts}

%\usepackage[OT4,OT1,T1]{fontenc}

%\usepackage[latin1,cp1255]{inputenc}

\usepackage{array}

\makeatletter

%\usepackage{babel}

%\newcommand{\theauthor}[1]{\hfill\begin{otherlanguage}{english}{\it #1}\end{otherlanguage} }

%\font\scf=ptmr8r scaled 910

%\font\kukua=ptmr8r scaled 1370

%\font\kukub=ptmr8r scaled 1200

%\usepackage{ij}

%\usepackage{times}

\newcommand{\rF}{\operatorname{F}}

\newcommand{\rG}{\operatorname{G}}

\newcommand{\rSp}{\operatorname{Sp}}

\newcommand{\bbR}{\mathbb{R}}





\def\ {\thinspace}

\lineskiplimit=100pt 

\lineskip=6.9pt



\def\index\indexvolume#1\indexyear#2\endindextitle{

\rm\parindent=0pt%\overfullrule=0pt

\centerline{ISRAEL JOURNAL OF MATHEMATICS}

\vskip0.4cm

\centerline{TABLE OF CONTENTS, VOLUME #1\unskip, #2\unskip}

\vskip1.2cm

\frenchspacing}

\def\endindex{}





%\def\indexitem\indexauthor#1\indextitle#2\indexpage#3\endindexitem

%{\halign {\normalbaselines%

%\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%

%&\hbox to 1.3cm{\hfill ##}\cr%

%{\sc #1\unskip}\quad{\sl #2} & #3 \cr}\par}

%\def\indexauthor{}

%\def\endindexitem{}

%\def\indexpage{}

%\def\indextitle{}

\newcommand{\indexitem}[3]{\halign {\normalbaselines%

\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%

&\hbox to 1.3cm{\hfill ##}\cr%

{\sc #1\unskip}\quad{\sl #2} & #3 \cr}\par}



%\def\indexsee\indexauthor#1\partner#2\endindexitem

%{{\sc #1\unskip\enspace}{\sl See }{\sc #2\unskip}\par}



\newcommand{\indexsee}[2]{{\sc #1\unskip\enspace}{\sl See }{\sc #2\unskip}\par}



\makeatother



\begin{document}

\pagestyle{empty}



\index

\indexvolume 159

\indexyear 2007

\endindextitle



%3377 1

\indexitem{Abraham, Uri and Rene Schipperus}{Infinite games on finite sets}{205}



%3376 2

\indexitem{Avil\'es, Antonio}{The number of weakly compact sets which generate a Banach space}{189}



%3411 3

\indexitem{Bernig, Andreas}{The normal cycle of a compact definable set}{373} 



%3388 4

\indexitem{Frauenfelder, Urs and Felix Schlenk}{Hamiltonian dynamics on convex symplectic manifolds}{1}



%3390 5

\indexitem{Howard, R.\ and D.\ Hug}{Smooth convex bodies with proportional projection functions}{317}



%3390 6

\indexsee{Hug, D.}{R.\ Howard}



%3379 7

\indexitem{Joswig, Michael and Thilo R{\"o}rig}{Neighborly cubical polytopes and spheres}{221}



%3391 8

\indexitem{Lee, Han Ju}{Complex convexity and monotonicity in quasi-Banach lattices}{57}



%3402 9

\indexitem{Lerner, Andrei K.}{$BMO$-boundedness of the maximal operator for arbitrary measures}{243}



%3397 10

\indexitem{Loke, H.\ Y.\ and G.\ Savin}{On local lifts from $\rG_2(\bbR)$ to $\rSp_6(\bbR)$ and $\rF_4(\bbR)$}{349}



%3375 11

\indexitem{Loukaki, Maria}{On distinct character degrees}{93}



%3392 12

\indexitem{Mattarei, Sandro}{Inverse-closed additive subgroups of fields}{343}



%3400 13

\indexitem{Mitra, Sudeb}{Extensions of holomorphic motions}{277}



%3379 14

\indexsee{R{\"o}rig, Thilo}{Michael Joswig}



%3394 15

\indexitem{Ros\l anowski, A.\ and S.\ Shelah}{Sheva-Sheva-Sheva: Large creatures}{109}



%3413 16

\indexitem{Rovenski, Vladimir}{Submanifolds with restrictions on extrinsic $q$-th scalar curvature}{289} 



%3397 17

\indexsee{Savin, G.}{Hung Yean Loke}



%3377 18

\indexsee{Schipperus, Rene}{Uri Abraham}



%3388 19

\indexsee{Schlenk, Felix}{Urs Frauenfelder}



%3385 20

\indexitem{Schraudner, M.}{The canonical-boundary representation for automorphism groups of locally compact countable state Markov shifts}{253}



%3394 21

\indexsee{Shelah, S.}{A.\ Ros\l anowski}



%3401 22

\indexsee{Wang, Y.}{H.\ Wei}



%3401 23

\indexitem{Wei, H.\ and Y.\ Wang}{On $CAS$-subgroups of finite groups}{175}





\end{document}